**ORIGINAL PAPER** 



# In the hands of users with intellectual disabilities: co-designing tangible user interfaces for mental wellbeing

Kieran Woodward<sup>1</sup> · Eiman Kanjo<sup>1,2</sup> · David J. Brown<sup>1</sup> · T.M. McGinnity<sup>1</sup> · Gordon Harold<sup>3</sup>

Received: 18 October 2022 / Accepted: 24 August 2023 / Published online: 3 October 2023 © The Author(s) 2023

#### Abstract

Involving and engaging people with intellectual disabilities on issues relating to their mental wellbeing is essential if relevant tools and solutions are to be developed. This research explores how inclusive and participatory co-design techniques and principles can be used to engage people with intellectual disabilities in designing innovations in mental wellbeing tangible technologies. In particular, individuals with intellectual disabilities participated in a co-design process via a series of workshops and focus groups to design tangible interfaces for mental wellbeing as their wellbeing challenges are often diagnostically overshadowed. The workshops helped participants explore new technologies, including sensors and feedback mechanisms that can help monitor and potentially improve mental wellbeing. The adopted co-design approach resulted in a range of effective and suitable interfaces being developed for varying ages.

# **1** Introduction

Traditional mental wellbeing assessment methods require people to be aware of their wellbeing status and seek help which can be challenging for those with intellectual disabilities [1]. Individuals with intellectual disabilities can often experience mental wellbeing challenges, but these are frequently overlooked and attributed erroneously to their disability (diagnostic overshadowing) or classified as challenging behaviour [2]. Furthermore, some individuals with intellectual disabilities face additional challenges in expressing their emotions, correctly interpreting social situations and predicting the behavioural consequences of specific actions [3–5]. These challenges often result in a significant impact on an individual's likelihood of engaging with mental healthcare systems.

 Kieran Woodward kieran.woodward@ntu.ac.uk
 Eiman Kanjo eiman.kanjo@ntu.ac.uk
 David J. Brown david.brown@ntu.ac.uk

- <sup>1</sup> Department of Computer Science, Nottingham Trent University, Nottingham, UK
- <sup>2</sup> Department of Computing, Imperial College London, London, UK
- <sup>3</sup> University of Cambridge, Cambridge, UK

Mental wellbeing is a state of wellbeing in which an individual realises his or her own abilities to cope with the normal stresses of life and can be impacted by emotions felt [6]. It is typically described as having either a positive or negative valence that, in contrast to emotions, is less specific, less intense and less likely to be provoked or instantiated by a particular stimulus or event [7]. Traditional methods used to assess mental wellbeing often use standardised clinical questionnaires, typically in the form of patient-reported outcome measures (PROMs) or experience sampling [8] to understand longitudinal variability. Self-reporting is used to enable people to record their emotions and stresses which can be assessed and monitored to help establish stressful triggers [9, 10]. However, self-reporting can take considerable time to assess as it must be completed over a long period to gain useful insights [11].

Tangible User Interfaces (TUIs) describe physical objects that are able to translate user actions into input events in the computer interface [12]. As TUIs are physical objects, they rely on users' knowledge of how the physical world works for interaction [13], making them more intuitive especially for people with less digital knowledge. Matthews and Doherty [14] and Niemantsverdriet and Versteeg [15] have reported that people are more likely to create stronger emotional attachments with physical devices such as TUIs rather than digital interfaces such as apps. Tangible devices provide a technological alternative to traditional self-reporting allowing users to report their current mental wellbeing in real time. Tangible manipulation presents an opportunity to develop novel devices, enabling unique interaction methods and encouraging continued engagement [16]. A technological solution that could actively monitor an individual's state of mental wellbeing and provide feedback could be extremely beneficial in improving accessibility to mental health tools for all [17].

Co-design, particularly in the context of tangible user interfaces (TUIs) for mental health, plays a crucial role in ensuring the accessibility and effectiveness of such technologies for individuals with intellectual disabilities. Co-design involves actively involving end users in the design process, recognising them as experts in their own experiences and needs. When applied to the development of TUIs for mental health assessment, co-design empowers individuals with intellectual disabilities to contribute their unique perspectives and insights, ultimately resulting in interfaces that are tailored to their specific requirements. For individuals with intellectual disabilities, traditional methods of mental health assessment often fall short in accurately capturing their experiences and challenges. Diagnostic overshadowing, where mental health symptoms are erroneously attributed to the disability itself, can lead to the underestimation or misinterpretation of mental wellbeing issues. Moreover, difficulties in expressing emotions and understanding social situations can further impede effective communication about mental health concerns. Involving individuals with intellectual disabilities in the co-design of TUIs helps address these challenges by allowing them to provide input on the design of interfaces and interaction methods that better align with their capabilities and communication preferences.

TUIs offer unique advantages in the context of mental health assessment for individuals with intellectual disabilities. As physical objects, TUIs leverage users' knowledge of the physical world, making interaction more intuitive and less reliant on digital literacy. This aspect of TUIs is particularly beneficial for individuals with limited digital knowledge or cognitive impairments. Co-design enables the identification and incorporation of tangible manipulation techniques that resonate with the target user group, fostering engagement and emotional attachment to the devices. By involving individuals with intellectual disabilities in the co-design process, TUIs can be customised to accommodate their specific sensory, cognitive and motor abilities, ensuring a more inclusive and accessible mental health assessment tool. Furthermore, co-design facilitates the exploration and integration of multimodal sensors within TUIs. This allows for a comprehensive assessment of mental wellbeing beyond self-reporting, which may be challenging for individuals with intellectual disabilities. By gathering data from various sensory modalities, such as heart rate, skin conductance and movement patterns, TUIs can capture a more holistic picture of an individual's mental state. The involvement of end users in the co-design process ensures that the selection and integration of these sensors align with the needs and preferences of individuals with intellectual disabilities, maximising the accuracy and relevance of the collected data.

Therefore, an iterative co-design process is adopted not to achieve one specific TUI design; instead, a range of different devices was expected to be proposed to suit the needs of all potential users including those who may experience a wide range of limitations. The co-design process aimed to investigate the following research questions:

- 1. How can the co-design process be adapted to best suit the needs of those with intellectual disabilities?
- 2. What are the optimal design guidelines for prototyping mental wellbeing tangible interfaces for people with intellectual disabilities?
- 3. Which technologies can be embedded within TUIs for real-world wellbeing data collection based on people with intellectual disabilities?

To ensure the proposed TUIs are effective and usable by individuals who experience mental wellbeing challenges, including those with intellectual disabilities, it is imperative to design the interfaces with end users. An iterative co-design process is proposed that has been adapted to enable the designing of interfaces and exploration of different sensing and feedback mechanisms with participants from the Nottingham Interactive Community for Education Research (NICER) group at Oak Field School, Nottingham, UK. The group is formed of adults with a range of intellectual disabilities who have a wide range of experience in evaluating assistive technologies, including virtual environments and serious games. In particular, they are interested in research and want to make sure research about intellectual disabilities conducted locally involves people with intellectual disabilities and is informed by their needs and wishes. The developed approach aimed to alleviate communication challenges faced by the target group and gather feedback that resulted in the development of TUIs for monitoring real-world mental wellbeing.

The remainder of this paper is organised as follows: Section 2 provides a literature review on co-design; Section 3 describes the participatory iterative design methodology used to design and develop the interfaces; Section 4 provides a discussion and answers the research questions; and Section 5 concludes and offers recommendations.

# 2 Background

Co-design is the methodology for actively engaging people directly involved in an issue, place or process in its design, allowing them to make a meaningful contribution to the design process [18–20]. Co-design enables the reduction of the gap in knowledge between end users and researchers, allowing non-designers to become equal members of the design team, ensuring designer subjectivity is removed and the technologies developed are suitable for the target population [21, 22]. During the process, design tools are used to empower all of the participants to facilitate a 'joint inquiry' where 'problem and solution co-evolve' [23]. Co-design brings many benefits to the design of the project by helping the researcher better understand the challenges faced by users and any potential solutions [24, 25].

Co-designing helps solve real-world problems by bringing together people from different backgrounds into the design process, resulting in more inclusive solutions. However, to work most effectively, it is important to select appropriate methods and ways of working which need to match the project being designed and the potential users' capabilities and limitations. The following co-design approaches have been shown to help empower participants by making things that are normally challenging to observe available as resources for design during the co-design process [26–28]:

- The use of visual, creative methods helps maintain engagement and promotes idea generation [29].
- Physically making things helps people to explore, verbalise, remember and imagine helping them to generate new solutions to problems they have faced and offer their opinions [22].
- Creating and telling stories helps put things into context and provides a central way of sharing and communicating. Story sharing can be visual, verbal or include role play [29].

Co-design can bept used to promote the inclusion of people with intellectual disabilities when designing new solutions by including their personal experiences, making them more likely to take ownership of the final outcome [30]. People with intellectual disabilities may face barriers such as communication challenges when being involved in the codesign of new assistive technologies, resulting in co-design techniques needing to be modified to fit with participants' abilities [31]. Previous research has explored methods to help adapt and engage people with intellectual disabilities in the co-design process. Brereton et al. [32] presented early iterations of a design, such as initial prototypes, to engage people with intellectual disabilities in co-design. Alternatively, an iterative design methodology can be used to promote continuous engagement for those with intellectual disabilities [33]. The unfinished design can also be a probe itself to promote discussion and idea generation, such as proposed in the analysis of non-finito products [34]. Finally, focus groups have been used as a methodology for reporting and validating the experience of co-designers with disabilities [35].

Literature reviews [36, 37] explored inclusive research concluding that it adds value when there is a distinctive contribution which only co-researchers with intellectual disabilities can make, when it highlights the contributions people with intellectual disabilities make and when it contributes to better lives for the wider population of people with intellectual disabilities. This shows that co-design is imperative when the research answers questions the authors could not otherwise answer, reaches participants in ways that the authors could not otherwise access, involves reflecting upon the insider cultural knowledge of people with intellectual disabilities. This research resulted in the following inclusive research guidelines:

- Aiming to contribute to social change, that helps to create a society, in which excluded groups belong and which aims to improve the quality of their lives.
- Being based on issues important to a group, which draws on their experience to inform the research process and outcomes.
- Exploring which aims to recognise, foster and communicate the contributions people with intellectual disabilities can make.
- Providing information which can be used by people with intellectual disabilities to campaign for change on behalf of others.
- Showing that those involved in co-design are 'standing with' those whose issues are being explored or investigated.

There has been limited researchpt exploring TUIs for people with ID. Gelsomini et al. explored reflex [38] which provided a number of educational activities where users' manipulated physical items, demonstrating the benefits of using TUIs for the development of cognitive and social skills. Similarly, other work has demonstrated the benefits of TUIs with children with ID. Children with intellectual disabilities played with a range of tangible interfaces to demonstrate the effectiveness of tangible interfaces for learning [39]. The work concluded that the most efficient paradigm is one with a clear mapping between physical objects and their meanings. Similarly, Polipo a 3D-printed smart toy co-designed with special-education professionals, offers numerous physical affordances and feedback in the form of lights, sounds and music [40]. Polipo was shown to improve fine motor skills in children with intellectual disabilities and encourage them to communicate. Similar benefits were also observed with Poma [41], a TUI designed to improve social and cognitive skills of children with autism spectrum disorder. Overall, while

co-designing with participants who have intellectual disabilities has been previously explored, little research has considered the co-design and development of tangible interfaces for mental wellbeing monitoring and real-time feedback.

#### 2.1 Co-designing with people with ID

While co-design has been utilised to develop solutions for people with intellectual disabilities, much research does not involve those users in the co-design process. It is imperative to co-design with people who have intellectual disabilities to ensure solutions take into account the lived experiences of those facing challenges in order to develop the most appropriate solution that can have real-world impact. While there have been limited co-design studies for TUIs, co-design has been used to engage individuals with cognitive disabilities in successfully designing a picture-based remote communication system, helping the participants move from merely being passive onlookers to active participants [42]. Similarly, co-design workshops have engaged people with assisted living needs to develop technologies and services for new care solutions [43], engaged autistic children in co-designing technologies [44], researched accessible apps and games with students with learning disabilities [45] and helped people with complex communication needs to express themselves through art therapy [46]. While involving users with intellectual disabilities in the design process produces additional challenges such as additional ethical considerations [47], it is imperative to ensure the solutions developed meet the needs of the potential users.

A card-based toolkit has been co-designed with people with intellectual disabilities for smart outdoor experiences [48]. This COBO toolkit integrates inspirational cards, interactive smart objects and multimedia contents to guide users during the conception of novel ideas. Participants played a crucial role during the process and introducing the elements; they contributed with very important feedback about the game dynamics and the interactive board prototype current research mainly highlights the benefits of TUIs for children [43]. Similarly, a mobile app has been co-designed to help people with ID use public transport using four specific elements to increase user engagement: a digital prototype, a non-finito feature, an inclusion of a proxy and a co-development opportunity [49]. These four elements confirmed the benefits of prototypes and showed that 30-min co-design sessions were insufficient.

Co-designing with people with ID for mental health has also been conducted to develop and implement creative methods to engage young people experiencing psychosis in an online resource for educational and wellbeing support [50]. Through workshops involving storytelling with emojis, relatable personas, card sorting activities and collaborative design, the research successfully engaged young people, their

families and clinicians in both hospital and community mental health settings. These creative co-design sessions, inspired by social media metaphors and tailored to their interests and moods, facilitated active engagement, drew out unique perspectives and ultimately helped develop solutions to effectively meet their needs. A competency-based methodology that was also influenced by social media served as a collaborative effort in designing a web application for imparting life skills [51]. This competency-based approach expands upon ability-based design principles and ensures that the devised designs are accessible to a diverse user base, while also acknowledging the necessity for additional assistance and customization for specific individuals. This approach fosters empowerment by harnessing the competencies of individuals with intellectual disabilities, while also allowing for the advancement and further development of skills.

A game-based learning tool has also been co-designed with young people with ID concluding that during the codesign process, it is necessary to provide clear instruction, be flexible to engage all participants, provide the ability to adapt content to best suit the needs of the participants and provide an emphasis on the consequences of choices [52]. Similarly, a digital skills education solution has been codesigned with people who have ID combining co-design with focus groups and user testing to encourage authentic engagement and inclusion of people with ID and accessibility needs throughout the entire design process [53].

Co-designing with children with ID has been conducted to design a serious game using a flexible approach allowing them to express their voice as user, informant, designer and tester [54]. The study found that when participants were asked to think of a game design, they struggled without existing prompts or prototypes as a concrete starting point. Equity on all aspects was not achieved consistently through the project, but the students' voices remained empirical for the final solution showing how each stakeholder can bring their own expertise to the task. This study shows an informal, flexible and practical co-design approach is required to promote idea generation and active participation throughout the entire session. Similar research [55] that co-designed with adolescents with autism spectrum disorder also highlighted the benefits of prototyping and conducting an iterative design process. Overall research shows that when participants' needs, preferences and desires are accounted for, participants with ID can make meaningful contributions to the design process [56].

# 3 Co-design methodology

We have adopted a participatory approach to co-design solutions for the monitoring of mental wellbeing. This enables the design of products and applications directly with the final users including those with intellectual disabilities in order to develop accessible products. Co-design helps create a sense of empowerment and a feeling of competence, which benefits the participants as they derive satisfaction and fun while feeling useful through their participation [57, 58]. We therefore involved adults with intellectual disabilities throughout the entire co-design process as outlined in Fig. 1 with the assumption that their inclusion will positively impact the quality of the final product in addition to their own experience.

This instantiation of an iterative design cycle took over 1 year in total. Each stage of the co-design process was conducted with the same primary researcher and an experienced facilitator with many years of experience in running co-design workshops. All participants were members of the NICER group with varying disabilities including Williams syndrome (a rare genetic disorder resulting in mild to moderate delays in their cognitive development) [59], Down syndrome [60] and autism [61], but no participants had significant motor skill impairments that would impact their participation. Their role in this process was not as research subjects, they were instead involved in identifying design opportunities relevant to their needs.

A person diagnosed with an intellectual disability [62] shows deficits in intellectual functioning such as reasoning or problem-solving and deficits and impairments in adaptive functioning such as communication or social skills, all during the developmental period before the age of 18 [63]. Intellectual disabilities can be divided into four levels: mild, moderate, severe and profound where for each level the person requires more support [64]. Co-designing with people with disabilities presents additional challenges-such as issues regarding the nature of their cognitive disability, including communicational and memory issues, which may challenge their full participation. However, it is imperative for the voices of end users to be heard, and many of the challenges can be overcome by developing a co-design methodology catering for the requirements of people with intellectual disabilities. The early integration of people with intellectual disabilities into the design processes aimed to prioritise their design decisions and needs. These insights serve as guides to a joint inquiry that seeks to address the challenges of developing TUIs to monitor affective state.

The co-design process used in the project is depicted in Fig. 1 adapted from [36, 37] where people with intellectual disabilities engaged in a partnership with researchers. The methodology alternates between focus groups to gather feedback during the preparation, co-analysis and validation phases and interactive co-design workshops completing the ideation and development phases. A novel aspect of this process is that it enabled feedback to be gathered at each stage of the development process including design and prototype development. The co-design process was adapted to include more demonstrations and interactivity to engage participants

with less general discussions. Furthermore, the preparation phase introduced participants to the concept of mental wellbeing, and the researchers ensured familiarity during the subsequent workshops and validation phases.

Overall, the co-design process was conducted in conjunction with an advisory panel of adults with intellectual disabilities, focusing on methodological adaptations and special supports developed to facilitate and ensure their participation. The co-design sessions were recorded for future analysis as granted by Nottingham Trent University, ethics application 18/19-43V2.

#### 3.1 Preparation phase focus group

At the beginning of the co-design process, an introductory preparation phase was completed to agree on the scope and aims and objectives of the project. Members of NICER along with teachers of young students with moderate to severe intellectual disabilities acted as an advisory panel with a researcher, experienced facilitator and education specialist leading the session. The focus group was conducted over 1 h, and notes of both verbal and non-verbal communication were recorded for future analysis by the experience facilitator.

An accessible introduction was completed using a presentation to introduce participants to the concept of TUIs and the possibility for them to automatically monitor mental wellbeing. This was completed by showing examples of existing interfaces such as Emoball [65] and Mood TUI [66] to help participants develop a concrete understanding of TUIs and allow them to gain greater knowledge of the devices to be developed. Explanations of hard-to-understand concepts were provided including discussions on emotions and the technologies themselves.

After the project had been introduced, participants were asked whether it was of interest to them and whether they would like to get involved in the subsequent co-design workshops and they agreed and volunteered their involvement. The NICER group is a significant stakeholder in this research, and they developed high expectations of TUIs that could have a positive impact for the intellectually disabled community.

### 3.2 Co-design workshop 1

The first co-design workshop aimed to explore various designs, technologies and requirements for mental wellbeing TUIs. The workshop was conducted over 4 h at Nottingham Trent University to strengthen the participants' roles as experts and posit them as co-researchers within a university setting and was video and audio recorded for future analysis. The workshop comprised of six participants: four males and two females who have previously been involved with multiple research projects and are experienced co-designers having helped co-design multiple serious games

Preparation	Co-design workshop 1	Prototyping	Co-analysis	Co-design workshop 2	Product Development	Validation
• Focus group with NICER to introduce the concept of TUIs for wellbeing	Workshop with subset of NICER to establish requirements for wellbeing interface and explore various designs and technologies	<ul> <li>Primary reseaercher develops initial prototypes using the feedback gathered from the co-design workshop</li> </ul>	• Focus group with NICER to gather feedback from the co-design workshop and the resultant prototypes	• Workshop with subset of NICER to refine initial interface designs and explore additional designs and technologies	<ul> <li>Primary researcher refines the interfaces using the additional feedback gathered</li> </ul>	• Focus group with NICER to provide the revised interfaces to participants to gather final feedback

Fig. 1 Stages of iterative co-design process

and have been involved in numerous research projects with the same research facilitator in the past. While members of NICER have experience in the co-design process helping them to make valuable contributions and have lived experiences of mental wellbeing challenges, the electronic elements of developing TUIs were new. Therefore, little preparation regarding general co-design practices was required, and the previous preparation phase along with the co-design introduction helped introduce the concept of TUIs and the potential electronics that could be utilised.

Table 1 shows the number (N) of participants and their characteristics including those who have Williams syndrome (WS), Down syndrome (DS) and autism along with information on their gender and level of intellectual disability (moderate or severe) as defined by their condition and previous evaluation by professional carers and teachers. A common challenge faced by all participants is their difficulty expressing themselves verbally.

A common issue with the development of mental wellbeing technological solutions is the lack of ethical considerations [67]. To ensure the co-design workshops were inclusive for participants with intellectual disabilities and caused no harm, all discussions were short and a range of interactive tasks were designed to increase engagement. During the workshop, it was ensured that all participants had a full understanding of the goal of the workshop and their role as a co-designer. The following five co-design methods used within the co-design workshops were designed to support participants' decisions, help researchers understand the participants requirements and act as conversational instruments based on established methods [68] and previous experience [69, 70].

#### 3.2.1 Introduction, explanation and demonstration

When conducting co-design workshops with participants who have intellectual disabilities, it is imperative to ensure all participants fully understand the goal of the workshop to improve communication. To ensure this, an experienced facilitator introduced the session by clearly explaining the concept of affective tangible interfaces. The concepts of mental wellbeing and emotions were then discussed to ensure all of the participants had an understanding of what wellbeing is and how it impacts them in order to facilitate discussions on solutions to help wellbeing monitoring. Finally, the requirement to develop a device that can monitor wellbeing was discussed to ensure all participants had a full and comprehensive understanding of the purpose of the workshop. Participants had the opportunity to engage in discussions about the mental wellbeing challenges they face as well as what they would require from a technological solution to monitor wellbeing, demonstrating their understanding of both the concept of mental wellbeing and the workshop aim.

 Table 1
 Co-design workshop 1 participant characteristics (age, gender and disability severity (moderate or severe)) for total number of participants (N), with Williams syndrome (WS), Down syndrome (DS) and autism

		Total ( <i>N</i> =6)	WS ( <i>N</i> =1)	DS (N=4)	Autism (N=1)
Mean age		36	39	39.75	18
(range)		(18–47)		(30–47)	
Gender N	Men	4	1	2	1
	Women	2	0	2	0
Level of intellectual disability N	Moderate	1	0	1	0
	Severe	5	1	3	1



Fig. 2 Original mental wellbeing interface prototype

Challenges can arise from communicating with participants and difficulties interpreting non-verbal interactions [71, 72]. Therefore, in the co-design processes, the adage 'show me don't tell me' [18] is often used, resulting in previously developed prototypes such as the cube shown in Fig. 2 being demonstrated to the participants. Concrete prototypes create opportunities for participants with intellectual disabilities to interact directly with the interfaces and understand the feasibility of developing new interfaces [73]. The prototypes embedded a range of sensors including 9 Degree of Freedom Inertial Measurement Units (9-DOF IMU) to measure motion, force-sensitive resistors (FSR) to measure touch and HR and EDA sensors to measure physiological changes.

#### 3.2.2 Storyboarding and drawing

The design of the interfaces was then explored. Previous work provides many useful strategies for engaging individuals with intellectual disabilities in the design process such as storyboards and pictures and avoiding open-ended questions [74–78]. To better support participants' way of self-expression, real-time storyboarding was completed where prompts were presented to participants to expand upon, promoting communication. During storyboarding, participants were able to discuss their opinions on the existing interfaces

previously demonstrated and share their ideas for new interfaces.

In co-design, methods are used to help participants 'say, do and make' [79]. This helps us deepen our engagement with people and strengthens the insights we are able to gather. Using this approach, participants were invited to draw their own interfaces using pen and paper to help promote ideas for new interfaces. This enabled the participants to creatively express their design ideas without the need to verbalise, which some people with intellectual disabilities can find challenging.

#### 3.2.3 Prioritising design requirement cards

The potential requirements of mental wellbeing tangible interfaces were also explored as it is imperative to understand what features users most require to ensure successful devices are developed. A card-based approach was used that enabled participants to prioritise the features they believed were most required. This approach was based on the generative research approach [80] to combine participatory exercises with verbal discussion during the creative idea generation phase. Similar card-based approaches have previously been used due to their accessibility, familiarity and tactile nature which can help promote communication [81]. Six cards were provided to each participant stating a requirement for tangible interfaces including: ease of use, makes me feel better, design, battery life, physical size and understands how I am feeling. Each of the six requirements was explained to the participants by the researcher and experienced facilitator to ensure they fully understood the meaning of the requirements and their role in prioritising the requirement cards.

Each requirement was given a score dependent on the order in which each participant placed the requirement, where the highest priority was given a score of six and the lowest one (Table 2). The most prioritised requirement was *makes me feel better* (22), closely followed by *ease of use* (21), then *battery life* (19), *size* (16), *understands how I feel* (14) and finally *design* (13). This shows that the participants all value the feedback the device could provide to make them feel better as the highest priority. However, this would

 Table 2
 Five participants' tangible interfaces priorities as ordered during the co-design workshop from highest to lowest priority

Participant 1	Participant 2	Participant 3	Participant 4	Participant 5
Ease of use	Battery life	Makes me feel better	Size	Battery life
Makes me feel better	Makes me feel better	Understands how I am feeling	Makes me feel better	Ease of use
Design	Ease of use	Ease of use	Understands how I am feeling	Size
Battery life	Design	Size	Battery life	Design
Size	Understands how I am feeling	Design	Ease of use	Understands how I am feeling
Understands how I am feeling	Size	Battery life	Design	Makes me feel better

first require the device to understand how the individual is feeling which was the second least prioritised feature, possibly showing a lack of understanding of the *understands how I am feeling* requirement. This is potentially due to this requirement being an essential prerequisite to the other options; therefore, participants may have considered this the core functionality rather than a design requirement. *Ease of use* was the second highest rated priority showing it to be highly valued amongst all participants. *Battery life* and *size* followed, although this greatly varies between participants, with some participants rating them as the highest priority and others rating them as the lowest priority. The lowest overall priority was *design* which was unexpected as participants enjoyed exploring the different prototypes suggesting they see *makes me feel better* as the overall design goal.

#### 3.2.4 Real-time 3D printing

The process of showing participants with an intellectual disability how their design decisions have a direct real-world consequence in a rapid and concrete way was developed in an earlier study and replicated here [82]. Three-dimensional modelling software was demonstrated by the researcher to demonstrate how the interfaces can be designed and printed to make concrete the relationship between the participants'



Fig. 3 Participants exploring different haptic patterns during a codesign workshop

decisions and the tangible interfaces produced. A majority vote was conducted to decide on the shape to be printed, and then participants were involved in collaboratively designing the shape using the software. As the workshops were conducted over several hours, there was sufficient time to design and 3D print a small interface, providing opportunity for participants to provide reflection on the design. Creating the interfaces during the workshops resulted in a deeper and more practical understanding about the participants' experiences [79].

#### 3.2.5 Interactive electronics

When exploring new technological solutions, providing demonstrations is necessary to ensure all participants understand the functionality and how the technology can be used, thereby improving confidence and communication [83]. During the session, a range of non-invasive, easy-to-use sensors were explored through interactive demonstrations. This was designed to increase engagement and ensure participants understood the functionality of the electronics by allowing them to experience the different capabilities offered by each sensor. All electronics were made simple to operate with the electrical circuits pre-built, as used in previous co-design studies [84], to ensure all participants would be able to fully participate [85].

A HR sensor was first explored where participants were able to place their finger on the sensor and lights would flash at the same rate as their pulse. An EDA sensor was also explored as it functions in a similar way to the HR sensor with participants having to place their fingers on the sensor. An FSR was demonstrated next, where as participants pressed harder on the sensor, it caused a haptic motor to vibrate. Finally, a 9-DOF IMU was demonstrated inside a ball; as participants shook the ball, it would vibrate. Overall, this method of exploring the sensors promoted participants' understanding and enabled them to experience how the sensors will be used in future interfaces.

Varying forms of feedback acting as real-time interventions were also explored giving participants time to reflect and express their feedback between demonstrations (Fig. 3). Visual feedback continued to be explored following the use of multicoloured LEDs to demonstrate the HR sensor. Participants were shown multiple examples including one device where different colours represented different emotions. Auditory feedback was also demonstrated where a speaker was used to play calming sounds from nature. Haptic feedback was the last intervention explored; four different feedback patterns were demonstrated to each of the participants who held the vibration motor to experience the different sensations.

#### 3.2.6 Exploration of tangible self-report methods

Methods to self-report were discussed, as in order to gain a real-world understanding of mental wellbeing, the sensor data along with the user's state of wellbeing is required. It is necessary to collect self-reports at the point of collection as it is challenging to identify afterwards. On-device self-reporting simplifies the process of by not requiring additional materials, such as questionnaires, and its ease of access promotes frequent reporting. During the session, a range of non-invasive, easy-to-use sensors and on-device self-reporting methods such as buttons and sliding potentiometers that could be used in real-world environments [86] were explored through interactive demonstrations.

A variety of buttons, sliding potentiometers and forcesensitive resistors were demonstrated as on-device methods to label wellbeing. Participants had the opportunity to test each of the labelling methods by simulating using the labelling methods to self-report their current state of mental wellbeing. During the discussion, all participants understood the need for labelling data and expressed their desire for on-device labelling over traditional paper-based labelling stating it is easier to use. Upon considering potential ondevice labelling methods, all participants preferred buttons stating I do think they would be better as they were the easiest to use, in particular using different buttons to represent different emotions. When considering the best methods to represent the buttons, Makaton was suggested which is a language programme that uses signs together with symbols to enable communication [87]. Alternatively, different coloured buttons to represent different emotions were explored, such as green and red buttons to represent positive and negative emotions respectively. Using different coloured buttons to label emotional states was preferred by all participants for its simplicity, but the number of labels would be limited to ensure simplicity for those with intellectual disabilities.

# 3.3 Prototyping

After the co-design workshop concluded, researchers developed initial high-fidelity prototypes using the designs, sensors and feedback mechanisms suggested. Participants were not involved in this process as it required technical skills such as soldering components and took several days, but their feedback during the co-design process guided all of the developed prototypes. This rapid prototype development was conducted to enable participants to physically experience their design suggestions from the workshop, promoting further discussion and the continued refinement of the interfaces (Fig. 4).



Fig. 4 Early prototyping with soft toys

### 3.4 Co-analysis focus group

During the co-analysis phase, focus groups were held where the participants who took part in the co-design workshop, along with other members of the NICER group and teachers of young students with intellectual disabilities, were provided with the opportunity to give feedback on the workshop and the resultant ideas. Typically, the focus groups were shorter in nature than the co-design session allowing for reflection on the outcomes of the workshops. During the co-analysis, notes of verbal and non-verbal communication were recorded for future analysis by the experienced facilitator.

The various activities conducted within the workshop were discussed to analyse what activities participants enjoyed and how they though the workshops could be improved, allowing for adjustments to be made to the co-design process where required. The group facilitator asked the NICER group members who attended the co-design workshop to present the main activities and outcomes of the workshop and their implications. Volunteers presented their memories, experiences and design preferences, and this again was all recorded using an accessible storyboard format. Implications for adjustment of co-design techniques and plans for follow-up activities at the next co-design workshop were also discussed.

#### 3.5 Co-design workshop 2

The second co-design workshop advanced upon the findings from the first workshop and co-analysis aiming to refine the developed interfaces. The workshop was conducted over 3 h and was audio recorded for future analysis. The workshop comprised of 8 participants: 5 males and 3 females who again are experienced co-designers. Five participants previously participated in the first co-design workshop; these included 1 with Williams syndrome and 4 with Down syndrome. Table 3 shows the characteristics of the participants including those

		Total ( <i>N</i> =8)	WS ( <i>N</i> =2)	DS ( <i>N</i> =4)	Other (N=2)
Mean age		36.5	38.5	39.75	28
(Range)		(27–47)	(38–39)	(30-47)	(27–29)
Gender $N(\%)$	Men	5 (62.5)	2 (100)	2 (50)	1 (50)
	Women	3 (37.5)	0 (0)	2 (50)	1 (50)
Level of intellectual disability N (%)	Moderate	1 (12.5)	0 (0)	1 (25)	0 (0)
	Severe	7 (87.5)	2 (100)	3 (75)	2 (100)

 Table 3
 Co-design workshop 2 participant characteristics including number of participants (N), with Williams syndrome (WS), Down syndrome (DS) and autism

who have Williams syndrome (WS) and Down syndrome (DS), along with information on their gender and level of intellectual disability.

During the co-design workshop, the five co-design methods previously developed were used. The session was introduced, and the previously developed interfaces were demonstrated to the participants, ensuring all participants were familiar with the interfaces and understood the purpose of the co-design session. After participants understood the aim of the session was to refine the existing prototypes, storyboarding was again utilised to help participants effectively communicate their new ideas for the interfaces. However, following feedback from the co-analysis and the challenges encountered during the first workshop where participants were not able to draw their own interfaces, this was not included within the second workshop. Instead, 3D modelling was performed, and participants experienced 3D printing a new interface during the workshop to help them understand the prototyping process and ensure they understood the impact their design decisions have on the developed interfaces.

The electronics within the interfaces were also explored with participants experiencing the same sensors as previously explored to ensure the suitability of the sensors and help gather additional feedback from new participants who had not previously experienced the electronics. Finally, potential features were not prioritised using cards as conducted in the initial workshop due to this workshop focusing on refining the ideas already produced. Instead, further discussions were held enabling participants to express their opinions on the required functionality and how the prototypes could be improved.

# 3.6 Product development

Using the feedback from the second co-design workshop, the initial prototypes were refined by the primary researcher. Similar to the prototyping stage, participants were not involved in this process as it required technical skills such as soldering components and took several days, but their feedback during the co-design process guided all of the developed devices. Multiple refined interfaces were developed over several weeks, each considering the feedback gathered throughout the iterative co-design process. These final products allow participants to experience how their design decisions impacted the development of solutions relevant to themselves.

Overall, nine prototypes were developed including both 3D-printed shapes and soft interfaces. The 3D-printed interfaces included 2 cubes as these are easy to hold, a cuboid, a sphere containing sleeves to place fingers within, a spheroid designed to ensure the user's thumb will rest on the HR sensor and their palm will touch the EDA sensor in addition to the torus shape selected to print during the second co-design workshop. The sensors within each device are shown in Table 4.

# 3.7 Validation focus group

The final aspect of the iterative co-design process was to evaluate the developed interfaces through a final focus group. Each of the developed interfaces was demonstrated to members of the NICER group including the participants from the co-design workshops and teachers of young students with intellectual disabilities where notes of verbal and non-verbal communication were taken by the experienced facilitator for future analysis. The participants had the opportunity to experience the different interfaces developed throughout the co-design process and examine how their feedback helped influence the design and functionality of the devices. Participants also had the opportunity to provide their final feedback on all concepts generated, including the design and self-report methods within the interfaces in addition to the co-design process itself and how it was adapted to promote communication and idea generation. This feedback can then be used to refine the developed TUIs as well as the co-design process for future studies.

During the evaluation, participants stated that the interfaces developed thus far are suitable for potential users. During the time, participants experienced the devices they stated that their use made them feel happy; in particular, the fidgeting buttons helped them feel calmer. Teachers and end

	and 1 possibuli of the 2 acteration wellocing 1 on and the	
Device	Image	Description
Ball		A soft ball embedding 9-DOF IMU to measure motion and capacitive sensor to measure touch with multi-coloured LEDs to perform visual feedback
Cube (touch)	0	A 3D-printed cube embedding 9-DOF IMU, capacitive touch, HR and EDA sensors with haptic feedback
Cube (buttons)	-	A 3D-printed cube embedding 9-DOF IMU, fidgeting buttons, HR and EDA sensors with haptic feedback
Teddy	0	A soft teddy embedding a 9-DOF IMU and capacitive touch sensor with visual feedback
Torus		A 3D-printed torus embedding HR, EDA, 9-DOF IMU and capacitive touch sensors with haptic feedback
Cushion		A soft cushion embedding EDA, 9-DOF IMU and capacitive touch sensors with haptic and visual feedback
Cuboid		A 3D-printed cuboid embedding 9-DOF IMU and capacitive touch sensors with visual feedback
Sphere		A 3D-printed sphere embedding HR, EDA and 9-DOF IMU sensors with haptic feedback
Pebble		A 3D-printed spheroid embedding HR, EDA and 9-DOF IMU sensors with haptic feedback

 Table 4
 Description of the 9 developed wellbeing TUIs and the embedded electronics

users agreed, liking the shapes of the interfaces in addition to the ubiquitous nature of the devices with the sensors being embedded within objects. Overall, the developed devices were found to be suitable for their intended purpose of monitoring mental wellbeing and collecting real-world data.

# **4** Discussion

This co-design study has explored the design, development and evaluation of TUIs to monitor real-world mental wellbeing. A co-design methodology was adopted based on previous research with participants whose mental wellbeing can often be diagnostically overshadowed and who commonly have difficulty in expressing their emotions [2]. The co-design approach addressed the limitations experienced by people with intellectual disabilities (e.g. communication, working memory), enabling them to participate more effectively. This approach takes a practical stance in guiding how co-design methods can be made to work in realistic design settings and adjusted to the needs of participants who may experience intellectual disabilities.

This research has many implications for both affective recognition technologies and the process of co-designing technologies with people who have intellectual disabilities. The five developed co-design methods aimed to increase individuals' autonomy [88] and promote communication within the

workshops. The participants' investment in the co-design process resulted from their ability to recognise the practical applications of mental wellbeing technologies while also appreciating the impact that they may have on the daily lives of diverse user populations.

When co-designing with people who have intellectual disabilities, it is vital to gradually unfold their creative potential to encourage meaningful participation [89]. Therefore, a number of activities were mediated with education professionals and conducted within each workshop that were designed to elicit design input and opinions, reach consensus and check understanding, such as the prioritisation of requirements.

A number of instructional and conversational instruments were also used during the co-design process to aid inclusivity [68]. These design instruments were used such as demonstrations and hands-on experiences with existing prototypes, sensors and actuators to serve as conversational instruments. During conversations, design instructions were iterated in conjunction with co-design participants, for example it was explained to participants that including all sensors in each design would lead to a TUI that was physically large. Furthermore, all information such as notes including non-verbal communication and video recording was collected in a structured way for subsequent analysis, and major outcomes were recorded on a note board in picture form and simple sentences as a joint record of achievement, which was easy to understand for the entire design team.

A range of interactive activities were also used as the co-design session lasted from morning to early afternoon, allowing them to incorporate physical (drawing, prioritising cards) and digital (3D modelling, exploration of electronics) elements and provide continuity between exploration and prototype testing. This meant that ideas captured from the whole team could be storyboarded and 3D printed all in 1 day, to make concrete the connection between design decisions and embodiment to support cognitive accessibility.

During the ideation phase, the production of prototypes was completed in a stimulating and playful environment, free from pressures that intimidate and block the creativity of the participants, including the use of design kits (drawing materials, cards, electronic components and sensors) as facilitating tools. The participants have previously been involved in the design of assistive technologies, and hence, they bring experience and expectations to be central to design decisions at co-design workshops.

These design instructions and activities along with the five developed co-design methods and the inclusion of analysis and validation phases ensured participants were able to fully participate and provide valuable feedback. The feedback gathered from the co-design process resulted in the development of multiple interfaces for real-world mental wellbeing monitoring.

#### 4.1 Research questions

The following sections use the analysis from the co-design process and feedback from the evaluation session where participants experienced the final developed interfaces to answer the four research questions.

# 4.1.1 How can the co-design process be adapted to best suit the needs of those with intellectual disabilities?

Co-designing for people with severe, profound and complex intellectual disabilities has shown additional challenges requiring adaptions to the co-design process to increase engagement. Not all tasks were successfully completed by all participants—such as drawing a new interface, as experienced in similar co-design studies [90]. However, the majority of the tasks were completed successfully, particularly the interactive activities within the co-design workshops, involving exploring sensors and feedback mechanisms and the 3D printing of new interfaces to immediately show participants the results of their design decisions. These interactive activities helped maintain engagement, improved understanding of new concepts and improved communication within the workshops. Additionally, storyboarding is an effective method for identifying, understanding and coming to grips with factors that capture and influence people's experiences [74–78]. A novel approach utilised was combining the interactive sessions with storyboarding to gain feedback, and it has been possible to gain valuable insights, aiding the design and development of future affective interfaces.

During the co-design workshops, prototypes were shown to participants to bridge the gap in knowledge from researchers to participants, helping them gain an understanding of the type of interfaces to be developed. However, to ensure participants' freedom was not restricted, a number of different prototypes were presented along with different materials and electronics. This helped demonstrate not only example prototypes but also the components used to develop such devices, allowing participants to have greater flexibility when devising their own suggestions. Previous co-design approaches with participants who have intellectual disabilities have similarly used prototypes to elicit feedback [91-93] and found it extremely beneficial. We experienced very similar results with participants becoming more engaged when using prototypes, evidenced by an increase in their attention and verbal feedback. Providing participants with low-fidelity prototypes offers much clearer expectation about the upcoming system and offers participants the opportunity to react to features. The use of prototypes and interactive sessions successfully aided idea generation during the co-design workshops while allowing participants to retain freedom in their designs.

Drawing new interfaces was also adapted to meet the needs of participants with ID by helping participants creatively express their ideas that they were not able to verbally communicate such as the sensors and feedback they believed were most important to include. The process of drawing an interface was a novel addition to the co-design process, and it was adapted from similar studies [79, 94, 95] that encourage participants to 'do' and 'make' although few co-design studies in particular that involved those with ID has provided the creative freedom to participants to envision their own interfaces without requiring verbal communication. Additionally, allowing participants to express themselves through 3D printing during the workshops helped demonstrate how the interfaces are developed and encouraged all participants to consider how differently shaped interfaces could be used. This process helped participants make a concrete connection between their design decisions (their drawings) and how this had a direct impact on the outcome (3D-printed interface) [82]. Linking the drawings from participants to a physical product (3D print) during the workshop is also a novel approach that worked extremely well. Allowing participants to move from the virtual environment to the physical world reinforced how their decisions had a tangible impact. While

previous work has developed low-fidelity prototypes during co-design sessions [96–98], the use of 3D printing and virtual modelling was a novel approach that goes beyond previous work to produce final products that better resemble what participants envisioned during the workshops.

Ranking features of the interfaces helped participants express what functionality they believed was most important without needing to verbally communicate. While this activity was based on the generative research approach [80, 81], we adapted the card-based approach to meet the needs of participants with ID by enabling them to rank features, thereby allowing time to reflect and make additional suggestions. Physically ranking features was a useful incorporation that has not been used in previous co-design studies with people with ID as it allows participants to consider what they most require while providing prompts to encourage further discussion. This is a novel approach that is often completed verbally during co-design sessions but by providing tangible requirements that participants physically ordered ensured they considered each of the requirements as evidenced through the subsequent discussions.

Finally, exploring the sensors and feedback was enjoyed by all, and the analysis showed it greatly improved attention and engagement in addition to aiding the understanding of each technology. Demonstrating how the electronics work aided participants' understanding of how the devices are developed and helped them make realistic suggestions regarding the design of future TUIs. While it is common during co-design workshops to explore the technologies being used, having hands-on experience with a range of demonstrations utilising various sensors and feedback actuators simply connected via wires in their raw form rather than within a complete prototype gave the participants a much deeper understanding of the technologies being utilised rather than them being abstracted within the TUI [48, 99] or simulated [100] as other co-design research has previously. While this approach of showing the raw sensors and electronics to participants is more complex to understand, it helped participants gain a greater understanding of what is required when developing a TUI and should not be abstracted away in future studies.

During the workshops, cognitive barriers were a theme that appeared throughout, mainly through lack of communication. Frequently, participants would require prompting when discussing specific topics, for example when explaining which sensors would be most appropriate for use. There was a large variation in the communication skills within the group, with some participants elaborating on their feedback in great detail and others who frequently replied with oneword answers, simply nod or shake their head or always agree with the other participants. These are challenging issues to overcome as some participants may not feel comfortable in expressing their opinions in groups. However, the involvement of an experienced facilitator who understood the commonly occurring communication issues for this group, the use of Makaton [87] where appropriate along with the interactive activities during the co-design workshops, helped improve communication with all participants. Additionally, providing initial low-fidelity prototypes for participants to explore aided their understanding of potential uses for the devices, and drawing an interface helped them to communicate new ideas and features. Furthermore, having the co-design process alternate between co-design workshops and feedback focus groups helped participants gain a clearer understanding of the interfaces and the technologies used within them, helping many to better communicate their ideas and hence provide feedback to help design an effective TUI. This alternating approach ensured participants' voices were heard during the co-design sessions and allowed them to provide subsequent feedback helping to continue refining the developed TUIs.

Overall, the interactive co-design approach adopted addressed the limitations experienced by people with intellectual disabilities (e.g. communication and working memory), enabling them to participate more effectively. The co-design approach was extremely successful in gathering the requirements for mental wellbeing TUIs mostly by helping concrete participants' ideas. The interactive activities conducted as part of the co-design workshops were extremely successful and should continue to be utilised in future studies to help participants generate ideas. This approach takes a practical stance in guiding how co-design methods can be made to work in realistic settings and adjusted to the needs of participants who experience intellectual disabilities.

# 4.1.2 What are the optimal design guidelines for prototyping mental wellbeing tangible interfaces for people with intellectual disabilities?)

1. Inconspicuous design During the workshops, numerous limitations were discussed, but all participants considered it highly important that the interfaces did not appear as a medical device in order to reduce stigma. When a participant previously used medical sensors, they stated *I felt awful, I was panicking...the first time I thought I'm not doing this.* This makes it vital that any physiological sensors and feedback mechanisms within the interfaces are non-invasive, easy to use and inconspicuous, so as to not induce additional stress. This shows similar results to other studies designing mental health technologies highlighting it as a priority [101]. As the sensors explored during the workshops were all small and unobtrusive, participants believed they were ideal to monitor real-world mental wellbeing. By developing wellbeing interfaces for the general population, as well as for those

experiencing mental health challenges, it will reduce the associated stigma by ensuring the devices are suitable for all.

The size of the cube prototype was liked by participants with them stating *like that size* when referring to the size of new interfaces. Participants discussed numerous usage scenarios for the portable interfaces including using them as work, college and at home. Participants envisioned carrying the device with them and possibly placing it on their desk allowing them to use it whenever they felt necessary to help them understand their state of mental wellbeing or help them relax. This shows the devices must remain small and inconspicuous [102].

The colour and personalisation of devices were repeatably mentioned by participants, and in particular, it was stated the devices should not be black and white and participants would like the ability for the devices to change different colours. When drawing future interfaces, a novel method to personalise the interfaces by attaching extensions that contain additional sensors or feedback was devised to increase engagement. The addition of extensions would enable the devices to adapt to the user enabling the most beneficial sensors and feedback to be included on an individual basis. Additionally, participants suggested the idea of a base interface being developed that can then be customised with different coloured cases to make the device more personal. This highlights the importance of design, while previous work has shown the value of colours, images and layouts in virtual environments [103], this research demonstrates how this translates to physical devices.

**2.** Tangible methods of physical interaction From the workshops, two main use cases emerged: the first being a tool for recording mental wellbeing state and the second being an interface that could serve as something that helps improve wellbeing. The second use case emerged as participants suggested that they would like toys for fidgeting, stroking, to have it play calming sounds or for recording messages. The most prioritised requirement was *makes me feel better*. This shows that the participants all value the feedback the device could provide to make them feel better as the highest priority.

When experiencing poor mental wellbeing, people often fidget with objects as fidgeting is a natural response that helps regulate stress [104, 105]. Previous research shows squeezing interactions are preferred by children when angry, but boredom was the most prevalent emotion to trigger fidgeting, and clicking was preferred when bored [106]. One might stroke a furry stuffed toy to calm down or because one is sad; therefore, a number of the co-designed interfaces included mechanisms for fidgeting whether that be stroking a soft toy or pressing buttons on a cube. This demonstrates that the fidgeting buttons enjoyed during the co-design workshops are a beneficial addition and shoulded within affective TUIs resulting in the development of tangible fidgeting interfaces highlighting the benefits of similar previous therapeutic robotics such as Paro [107].

**3.** Sensors for objective measurement In this work, participants also highlighted the benefits of objective measurement over continuous self-reporting. People with intellectual disabilities can find it challenging to express their emotions; therefore, TUIs to monitor wellbeing may help them better manage their emotions by automatically providing onboard interventions such as haptic feedback or calming sounds. This highlights the importance of going beyond self-reporting wellbeing TUIs as used in many previous studies [66, 108, 109] as the inclusion of physiological sensors can realise the potential for TUIs to provide unobtrusive objective monitoring and automated interventional feedback [110] to improve mental wellbeing.

Unlike many existing TUIs for mental wellbeing [66, 108, 109] which only allow for self-reporting, participants saw the benefits of including sensors to measure physiological and motion data to enable the development of future computational models. While there is a danger of relying on sensors to monitor mental wellbeing as it is inherently personal, advances in artificial intelligence are creating highly precise affective models that could help to greatly benefit populations that are traditionally underrepresented. Collecting physiological sensor data opens up new possibilities for objective data analysis compared with online systems that have previously been co-designed for people with ID [50, 51]. While there remain limitations with deep learning affective modelling such as the models only being as good as the data used to train them, this emphasises the importance of including participants with intellectual disabilities in studies.

**4. Privacy preservation** Privacy was a key consideration when developing the interfaces especially as they can measure physiological data that links to an individual's health. Therefore, the interfaces do not communicate with any external devices as all data processing and storage is local. Additionally, microphones were not included within the interfaces to capture audio data as continuous recording of voice would be required to infer wellbeing which may raise privacy concerns due to the highly sensitive nature of the data [111].

A potential alternative to inferring wellbeing from audio data would be to enable participants to verbally record how they are feeling, but this requires participants to continuously understand and report on their wellbeing similar to a self-report diary, rather than automatically and objectively inferring wellbeing which could be of much greater benefit. While the possibility of embedding a microphone for participants to record messages was explored, it was not considered a suitable or valuable method of data collection. Participants never expressed interest in using the microphone for recording and self-reporting, and there are numerous privacy concerns when recording people especially in real-world environments. Furthermore, participants with intellectual disabilities often find it challenging to communicate their wellbeing resulting in little benefit of embedding microphone.

Preserving privacy is a key design consideration that must be considered throughout the design process. TUIs offer many opportunities to collect objective sensor data which could be greatly beneficial, but this must be carefully paired with high levels of privacy to protect users and gain trust.

**5.** Age appropriateness The results from this study show there is no one-size-fits-all solution to the design of mental wellbeing interfaces. Instead, different devices should be developed for different age ranges with soft devices for young children and 3D-printed interfaces for older children and adults. Participants found that not all sensors would be appropriate for children, such as the HR sensor, as participants believed from past experiences that children would not be able to keep their fingers in continuous contact with the sensor for the device to accurately measure physiological changes. It was suggested that physiological sensors should be reserved for the 3D-printed interfaces and not the soft interfaces, where children in particular may continuously move their hands preventing accurate readings from being recorded.

A number of prototypes were developed and categorised into soft toys potentially for young children aged 5–8 (middle childhood) as this is when children develop relevant social, emotional and cognitive skills [112, 113] and 3D-printed interfaces for older children (8+) and adults. Age is an important factor to consider when developing tangible interfaces to ensure they are accessible and engaging, helping to reduce stigma with mental wellbeing tools.

Even though all of the participants in the workshops were over the age of 18 and suggesting ideas for children's interfaces, they were all drawing off personal experiences they experienced during childhood such as not being confident to express their emotions, resulting in the devices that can measure touch and motion promoting gentle interactions such as stroking. Furthermore, one participant only recently turned 18 allowing them to base their contributions for children's interfaces from recent past experiences, and some other participants are involved in working within a school for children with intellectual disabilities allowing them to have personal insight into the features required in wellbeing TUIs for children. The developed interfaces were also shown to teachers of children with intellectual disabilities who agreed with the requirements established during the co-design sessions and believed the resultant devices would be highly beneficial for children. Similarly, the 3D-printed interfaces were well

received by all including teachers and highlight the capability to collect real-world physiological data in addition to enabling fidgeting interactions. This shows the need for separate devices to target different populations and demonstrates the success of the co-design workshops.

Overall, participants suggested numerous outputs for the design of future devices. Based upon user feedback, haptic feedback is a key intervention that should continue to be explored as all participants found it relaxing and preferred similar vibration patterns. Visual, haptic and auditory feedback can all be embedded within future devices, although shape-shifting may be more challenging and the addition of a screen would require careful consideration to ensure, it does not induce further stress. Overall, participants believed by including these feedback mechanisms, future interfaces could improve their mental wellbeing in real time.

# 4.1.3 Which technologies can be embedded within TUIs for real-world wellbeing data collection based on people with intellectual disabilities?

TUIs possess many opportunities when combined with sensing technologies. This co-design study has aimed to identify the most appreciated technologies and sensors that can be embedded within TUIs to help automate mental wellbeing detection including physiological and motion sensors. This is a current issue as advances in AI are helping to automate wellbeing recognition [114–116], but there has been a lack of research exploring the real-world devices where this could be used, especially for those with ID.

Previous TUIs for mental wellbeing do not embed physiological sensors and have simply been used to enable the self-reporting of the current state of wellbeing using interaction techniques such as recording emotions by squeezing an electronic ball [65, 117]. While these devices only allow users to report a limited number of emotions, participants believed mental wellbeing and education were the areas where these devices could be of most use. Similarly, Mood Sprite [108] and Subtle Stone [109] are handheld devices that allow users to record their emotions. The devices record the time users create new sprites allowing them to be revisited much like a diary, again showing ways in which TUIs can accompany traditional techniques to make treatment more accessible and user-centric. However, these approaches show that there has been little consideration of sensor-based TUIs for wellbeing. Furthermore, there is no evidence of wellbeing TUIs being co-designed with people with ID even though they could offer great benefit.

In this co-design study, a range of non-invasive sensors were explored using co-design sessions to measure physiological changes and physical interactions. Physiological sensors measuring HR, HRV and EDA present the greatest opportunity to automatically monitor wellbeing due to their correlation with the sympathetic nervous system and their non-invasive nature [118-120]. However, the results from the co-design workshops show touch and motion sensors should also be included within all devices due to their simplicity and ease of use, with the more complex physiological sensors reserved for 3D-printed interfaces. Accelerometers could be easily embedded within all interfaces, with participants believing the manner in which the devices will be interacted with will be different depending on the user's state of wellbeing. Similarly, the ability to measure touch is useful as participants believed stroking the soft interfaces was relaxing as it simulated stroking a pet. Participants enjoyed interacting with touch to activate the feedback, such as pressing hard to enable the visual and haptic feedback as they found this method of interaction intuitive. The way in which the device was touched was also suggested as a mechanism to indicate wellbeing, with users potentially squeezing the device harder when angry. This shows similar results to previous therapy robots [107], but combining these sensors with physiological sensors offers many new opportunities for objective measurements. Overall, the physiological, motion and touch sensors included within the developed interfaces were considered beneficial to objectively measure changes in wellbeing and suitable for the target audience.

A range of feedback actuators to serve as interventions and improve wellbeing were explored including haptic, visual and auditory feedback in addition to fidgeting tools. The co-design workshops helped establish the requirements such as the preference for fidgeting buttons [104, 105] and slow haptic feedback [121] which have been shown to improve wellbeing. Participants believed these feedback mechanisms have great potential to improve wellbeing with some participants finding the fidgeting aspects of the final tangible fidgeting interfaces especially relaxing. This demonstrates the use of the device and the fidgeting buttons themselves can serve as a benefit of using the artefact without the need for additional feedback mechanisms. Visual feedback patterns were also enjoyed with participants finding them engaging and relaxing, but it was found that it is not possible to use light patterns or colours to convey information such as current emotional state to users with intellectual disabilities. Previous work has similarly found that directly informing users of their current state of wellbeing can have a detrimental impact [122]. Therefore, to ensure no additional stress is induced, visual feedback has only been used to display varying patterns to aid relaxation. Finally, while the use of auditory feedback was suggested to play stories and calming sounds, the quality of the sound is of high priority. Visual feedback, slow haptic feedback and fidgeting buttons were found most relaxing and have been embedded within the developed prototypes to act as real-time interventions.

The key design proposals discussed within the co-design workshops including appearance, materials and sensors were all similar to previous research that aimed to co-design TUIs for emotion recognition albeit not for participants with intellectual disabilities [102]. Jingar and Lindgren [102] similarly suggested methods of inferring emotions through touch and motion as suggested for the soft interfaces in this co-design study but did not consider the inclusion of physiological sensors which could have a major impact on the accurate monitoring of real-world wellbeing. There remain limitations in the use of TUIs with physiological sensors as users have to place their fingers on the sensors to record data, unlike alternative devices such as wearables where there is constant connection with the body, but the co-designed TUIs offer novel methods of interaction such as fidgeting which is not capable within wearables, allowing the same device to serve for both monitoring and improving wellbeing.

Overall, we contribute to the body of literature by demonstrating the possibility for a range of sensors and feedback actuators to be embedded within TUIs for mental wellbeing going beyond that of previous research which has mostly focused on tangible methods to self-report without objective data collection [66]. The ability to collect objective data increases the accuracy of wellbeing reporting as previous self-reporting TUIs have shown only a moderate agreement between the expected emotion and the recorded emotion [123].

# **5** Conclusion

The automatic inference of mental wellbeing would be extremely valuable for individuals with intellectual disabilities as their wellbeing is often diagnostically overshadowed and they can find it challenging to express their emotions. TUIs enable the necessary microprocessor and sensors required to monitor mental wellbeing to be embedded, in addition to providing feedback mechanisms that may help improve mental wellbeing in real time.

Inclusive co-design workshops and focus groups have been conducted to rethink the user design approach of mental wellbeing TUIs. Adjustments to traditional co-design techniques included demonstrations, real-time 3D printing, prioritising cards and interactive electronics to enable successful and practical co-design with people with intellectual disabilities. In particular, the 3D printing of new interfaces improved engagement, ensured participants understood the discussed technologies and demonstrated how their decisions influenced the final designs. Thematic analysis of the qualitative data outlined many recommendations and resulted in a range of new interfaces being developed. Overall, the participatory process has enabled the successful design and development of mental wellbeing TUIs for people with a range of intellectual disabilities.

# Declarations

Conflict of interest The authors declare no competing interests.

**Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecomm ons.org/licenses/by/4.0/.

# References

- Clement S, Schauman O, Graham T, Maggioni F, Evans-Lacko S, Bezborodovs N, Morgan C, Rüsch N, Brown JSL, Thornicroft G (2015) What is the impact of mental health-related stigma on help-seeking? A systematic review of quantitative and qualitative studies. Psycho Med 45(01):11–27
- 2. Giraud-Saunders A (2011) Mental health in people with learning disabilities
- Adams D, Oliver C (2011) The expression and assessment of emotions and internal states in individuals with severe or profound intellectual disabilities. https://doi.org/10.1016/j.cpr.2011. 01.003
- Bruno RM (1981) Interpretation of pictorially presented social situations by learning disabled and normal children. J Learn Disabilities. https://doi.org/10.1177/002221948101400617
- Saloner MR, Gettinger M (1985) Social inference skills in learning disabled and nondisabled children. Psycho Schools. https://doi.org/10.1002/1520-6807(198504)22:2<201::AID-PITS2310220214>3.0.CO;2-V
- WHO (2014) World Health Organisation: WHO | Mental health: a state of well-being. World Health Organization. https://www.who. int/features/factfiles/mental\_health/en/ Accessed 11 June 2018
- Boraud T (2020) How the brain makes. Decisions. https://doi.org/ 10.1093/oso/9780198824367.001.0001
- Myin-Germeys I, Kasanova Z, Vaessen T, Vachon H, Kirtley O, Viechtbauer W, Reininghaus U (2018) Experience sampling methodology in mental health research: new insights and technical developments. World Psychiatry
- Stone AA, Turkkan JS, Bachrach CA, Jobe JB, Kurtzman HS, Cain VS (eds) (2000) The science of self-report: implications for research and practice. Lawrence Erlbaum Associates Publishers, Mahwah, pp –380380
- 10. Gustavson JG, R A (1997) The science of self-report. APS Observer 10(1)

- Stone AA, Shiffman S, Schwartz JE, Broderick JE, Hufford MR (2002) Patient non-compliance with paper diaries. BMJ (Clinical research ed.) 324(7347):1193–1194
- Picard RW (1997) Affective computing. http://www.media.mit. edu/~picard/
- Jacob RJK, Shaer O, Girouard A, Hirshfield LM, Horn MS, Solovey ET, Zigelbaum J (2008) Reality-based interaction: a framework for post-WIMP interfaces. In: Conference on human factors in computing systems - proceedings. https://doi.org/10. 1145/1357054.1357089
- 14. Matthews M, Doherty G (2011) In the mood: engaging teenagers in psychotherapy using mobile phones. Proceedings of the 2011 annual conference on Human factors in computing systems - CHI '11
- Niemantsverdriet K, Versteeg M (2016) Interactive jewellery as memory cue. In: Proceedings of the TEI '16: tenth international conference on tangible, embedded, and embodied interaction -TEI '16. ACM Press, New York, pp 532–538. https://doi.org/10. 1145/2839462.2856524
- Hornecker E, Buur J (2006) Getting a grip on tangible interaction: a framework on physical space and social interaction. Proceedings of the SIGCHI conference on Human Factors in computing systems - CHI '06. https://doi.org/10.1145/?????????
- Calvo RA, Peters D (2013) Promoting psychological wellbeing: loftier goals for new technologies [opinion]. https://doi.org/10. 1109/MTS.2013.2286429
- Burkett I (2016) An introduction to co-design. Technical report. http://www.csi.edu.au/
- Binder T, De Michelis G, Ehn P, Jacucci G, Linde P, Wagner I (2019) Participation in design things. In: Design things. https:// doi.org/10.7551/mitpress/8262.003.0011
- 20. Holmlid S (2009) Participative, co-operative, emancipatory: from participatory design to service design. First Nordic Conference on Service Design and Service Innovation
- Vines J, Clarke R, Wright P, McCarthy J, Olivier P (2013) Configuring participation: on how we involve people in design. In: Conference on human factors in computing systems - proceedings. https://doi.org/10.1145/2470654.2470716
- 22. Sanders L (2008) An evolving map of design practice and design research. Interactions. https://doi.org/10.1145/1409040.1409043
- 23. Steen M (2013) Co-design as a process of joint inquiry and imagination. Design Issues. https://doi.org/10.1162/DESI\_a\_00207
- 24. Steen M, Manschot M, de Koning N (2011) Benefits of co-design in service design projects. Inter J Design
- Skliarova I, Sklyarov V (2019) Hardware/software co-design. In: Lecture notes in electrical engineering. https://doi.org/10.1007/ 978-3-030-20721-2\_6
- Visser FS (2009) Bringing the everyday life of people into design. Isbn 978-90-9024244-6
- Edvardsson B, Meiren T, Schäfer A, Witell L (2013) Having a strategy for new service development - does it really matter? J Service Manage. https://doi.org/10.1108/09564231311304170
- von Hippel E (1994) Sticky information and the locus of problem solving: implications for innovation. Manage Sci. https://doi.org/ 10.1287/mnsc.40.4.429
- 29. Sanders E (2006) Design research in 2006. Design Research Society
- Sarmiento-Pelayo MP (2015) Co-design: a central approach to the inclusion of people with disabilities. Revista Facultad de Medicina. https://doi.org/10.15446/revfacmed.v63n3sup.49345
- Boot FH, Owuor J, Dinsmore J, Maclachlan M (2018). Access to assistive technology for people with intellectual disabilities: a systematic review to identify barriers and facilitators. https://doi. org/10.1111/jir.12532
- Brereton M, Sitbon L, Abdullah MHL, Vanderberg M, Koplick S (2015) Design after design to bridge between people living

🖄 Springer

with cognitive or sensory impairments, their friends and proxies. 11(1):4–20. https://doi.org/10.1080/15710882.2015.1009471

- Luckin R, Dunckley L, Dearden AM (2007) Designing human centred technologies for the developing world: HCI but not as we know it. People and computers XXI HCI.But not as we know it - proceedings of HCI 2007: the 21st British HCI group annual conference 2. https://doi.org/10.14236/EWIC/HCI2007.96
- Seok JM, Woo JB, Lim YK (2014) Non-finito products: a new design space of user creativity for personal user experience. Conference on human factors in computing systems - proceedings, pp 693–702. https://doi.org/10.1145/2556288.2557222
- Kidney CA, McDonald KE (2014) A toolkit for accessible and respectful engagement in research. 29(7):1013–1030. https://doi. org/10.1080/09687599.2014.902357
- Walmsley J, Strnadová I, Johnson K (2018) The added value of inclusive research. J Appl Res Intellect Disabilities. https://doi. org/10.1111/jar.12431
- Walmsley J (2004) Inclusive learning disability research: the (nondisabled) researcher's role. British J Learn Disabilities. https://doi.org/10.1111/j.1468-3156.2004.00281.x
- Gelsomini M (2018) Reflex: learning beyond the screen in a simple, fun and affordable way. Conference on human factors in computing systems - proceedings. https://doi.org/10.1145/ 3170427.3180302
- Falcão TP, Price S (2012) Tangibles for students with intellectual disabilities. ACM international conference proceeding series, pp 371–374. https://doi.org/10.1145/2307096.2307172
- Tam V, Gelsomini M, Garzotto F (2017) Polipo a tangible toy for children with neurodevelopmental disorders. TEI 2017 - proceedings of the 11th international conference on tangible, embedded, and embodied interaction, pp 11–20. https://doi.org/ 10.1145/3024969.3025006
- Al Mahmud A, Soysa AI (2020) POMA: a tangible user interface to improve social and cognitive skills of Sri Lankan children with ASD. Inter J Human-Comput Studies 144. https://doi.org/ 10.1016/J.IJHCS.2020.102486
- 42. Dawe M (2007) Let me show you what i want: engaging individuals with cognitive disabilities and their families in design. In: Conference on human factors in computing systems - proceedings. https://doi.org/10.1145/1240866.1240976
- 43. Wherton J, Sugarhood P, Procter R, Hinder S, Greenhalgh T (2015) Co-production in practice: how people with assisted living needs can help design and evolve technologies and services. Implement Sci. https://doi.org/10.1186/s13012-015-0271-8
- 44. Spiel K (2020) SIGCHI outstanding dissertation award: evaluating experiences of autistic children with technologies in co-design. In: Conference on human factors in computing systems - proceedings. Association for Computing Machinery, New York, pp 1–4. https://doi.org/10.1145/3334480.3386151
- 45. Anthony L, Prasad S, Hurst A, Kuber R (2012) A participatory design workshop on accessible apps and games with students with learning differences. In: ASSETS'12 - proceedings of the 14th international ACM SIGACCESS conference on computers and accessibility. https://doi.org/10.1145/2384916.2384979
- Lazar A, Feuston JL, Edasis C, Piper AM (2018) Making as expression: informing design with people with complex communication needs through art therapy. In: Conference on human factors in computing systems - proceedings. https://doi.org/10. 1145/3173574.3173925
- Spiel K, Brulé E, Frauenberger C, Bailly G, Fitzpatrick G (2018) Micro-ethics for participatory design with marginalised children. https://doi.org/10.1145/3210586.3210603
- Cosentino G, Morra D, Gelsomini M, Matera M, Mores M (2021) COBO: a card-based toolkit for co-designing smart outdoor experiences with people with intellectual disability. Lecture notes

in computer science (including subseries lecture notes in artificial intelligence and lecture notes in bioinformatics) 12932 LNCS, pp 149–169. https://doi.org/10.1007/978-3-030-85623-6 11/FIGURES/8

- Sitbon L, Farhin S (2017) Co-designing interactive applications with adults with intellectual disability: a case study. ACM Inter Conf Proc Series 487–491. https://doi.org/10.1145/3152771. 3156163
- Nakarada-Kordic I, Hayes N, Reay SD, Corbet C, Chan A (2017) Co-designing for mental health: creative methods to engage young people experiencing psychosis. Design Health 1(2):229– 244. https://doi.org/10.1080/24735132.2017.1386954
- Bayor AA, Brereton M, Sitbon L, Ploderer B, Bircanin F, Favre B, Koplick S (2021) Toward a competency-based approach to co-designing technologies with people with intellectual disability. ACM Trans Access Comput 14(2). https://doi.org/10.1145/ 3450355
- Raman S, French T (2021) Enabling genuine participation in co-design with young people with learning disabilities. Inter J CoCreation Design Arts 18(4):431–447. https://doi.org/10.1080/ 15710882.2021.1877728
- 53. Murphy E, Shiels O, Yap ML, Angus FC, Delaney J, Hogan B, Burke C, Heduan D, Breslin D, Fiori S (2022) Feel proud that with this app we can help other people with intellectual disabilities: the role of an expert by experience digital inclusion citizen advisory panel. Disabilities 2(4):715–735. https://doi.org/ 10.3390/DISABILITIES2040050
- Bossavit B, Parsons S (2016) This is how I want to learn: high functioning autistic teens co-designing a serious game. Conference on human factors in computing systems - proceedings, pp 1294–1299. https://doi.org/10.1145/2858036.2858322
- Zhu R, Hardy D, Myers T (2019) Co-designing with adolescents with autism spectrum disorder: from ideation to implementation. ACM Inter Conf Proc Series 106–116. https://doi.org/10.1145/ 3369457.3370914
- Fekete G, Lucero A (2019) P(L)AY ATTENTION! Co-designing for and with children with attention deficit hyperactivity disorder (ADHD). Lecture notes in computer science (including subseries lecture notes in artificial intelligence and lecture notes in bioinformatics) 11746 LNCS, pp 368–386. https://doi.org/10.1007/978-3-030-29381-9\_23/FIGURES/10
- Frauenberger C, Good J, Alcorn A, Pain H (2012) Supporting the design contributions of children with autism spectrum conditions. ACM Inter Conf Proc Series 134–143. https://doi.org/10.1145/ 2307096.2307112
- Malinverni L, Mora-Guiard J, Padillo V, Mairena MA, Hervás A, Pares N (2014) Participatory design strategies to enhance the creative contribution of children with special needs. In: ACM international conference proceeding series. https://doi.org/ 10.1145/2593968.2593981
- Mervis CB, Robinson BF, Bertrand J, Morris CA, Klein-Tasman BP, Armstrong SC (2000) The Williams syndrome cognitive profile. Brain Cognit 44(3):604–628. https://doi.org/10.1006/BRCG. 2000.1232
- 60. Silverman W (2007) Down syndrome: cognitive phenotype. Mental Retardation Develop Disabilities Res Rev 13(3):228–236. https://doi.org/10.1002/MRDD.20156
- Baron-Cohen S (1988) Social and pragmatic deficits in autism: cognitive or affective? J Autism Develop Disorders 18(3):379– 402. https://doi.org/10.1007/BF02212194/METRICS
- 62. Schalock RL, Borthwick-Duffy SA, Bradley VJ, Buntinx WHE, Coulter DL, Craig EM, Gomez SC, Lachapelle Y, Luckasson R, Reeve A, Shogren KA, Snell ME, Spreat S, Tasse MJ, Thompson JR, Verdugo-Alonso MA, Wehmeyer ML, Yeager MH (2010) Intellectual disability: definition, classification, and systems of supports. Eleventh Edition

- American Psychiatric Association (2013) DSM-5 diagnostic classification. In: Diagnostic and statistical manual of mental disorders. https://doi.org/10.1176/appi.books.9780890425596. x00diagnosticclassification
- 64. Luckasson R, Borthwick-Duffy S, Buntinx WHE, Coulter DL, Craig EMP, Reeve A, Schalock RL, Snell ME, Spitalnik DM, Spreat S, Tassé MJ (2002) Mental retardation: definition, classification, and systems of supports, 10 edn. American Association on Mental Retardation, ???
- 65. Fuentes C, Rodríguez I, Herskovic V (2015) EmoBall: a study on a tangible interface to self-report emotional information considering digital competences. In: Lecture notes in computer science (including subseries lecture notes in artificial intelligence and lecture notes in bioinformatics). https://doi.org/10.1007/978-3-319-26508-7\_19
- Sarzotti F (2018) Self-monitoring of emotions and mood using a tangible approach. Comput 7(1):7
- 67. Sanches P, Janson A, Karpashevich P, Nadal C, Qu C, Roquet CD, Umair M, Windlin C, Doherty G, Höök K, Sas C (2019) HCI and affective health taking stock of a decade of studies and charting future research directions. In: Conference on human factors in computing systems proceedings. Association for Computing Machinery, ???, p 17. https://doi.org/10.1145/3290605.3300475
- Spencer González H, Vega Córdova V, Exss Cid K, Jarpa Azagra M, Álvarez-Aguado I (2020) Including intellectual disability in participatory design processes: methodological adaptations and supports. In: ACM international conference proceeding series. https://doi.org/10.1145/3385010.3385023
- Brown DJ, McHugh D, Standen P, Evett L, Shopland N, Battersby S (2011) Designing location-based learning experiences for people with intellectual disabilities and additional sensory impairments. Comput Education. https://doi.org/10.1016/j.compedu. 2010.04.014
- Brown D, Kettley R, Kettley S (2016) Interpersonal process recall

   a method for reflecting on co-design experiences. In: Method stories workshop, ACM CHI. https://chimethodstories.wordpress.com/
- Hendriks N, Slegers K, Duysburgh P (2015) Codesign with people living with cognitive or sensory impairments: a case for method stories and uniqueness. Int J CoCreation Design Arts
- O'Connor C, Fitzpatrick G, Buchannan-Dick M, McKeown J (2006) Exploratory prototypes for video: interpreting PD for a complexly disabled participant. In: ACM international conference proceeding series. https://doi.org/10.1145/1182475.1182500
- Brereton M, Sitbon L, Abdullah MHL, Vanderberg M, Koplick S (2015) Design after design to bridge between people living with cognitive or sensory impairments, their friends and proxies. CoDesign. https://doi.org/10.1080/15710882.2015.1009471
- 74. Boyd-Graber J, Nikolova S, Moffatt K, Kin K, Lee J, Mackey L, Tremaine M, Klawe M (2006) Participatory design with proxies: developing a desktop-PDA system to support people with aphasia. In: Conference on human factors in computing systems - proceedings
- Galliers J, Wilson S, Roper A, Cocks N, Marshall J, Muscroft S, Pring T (2012) Words are not enough: empowering people with aphasia in the design process. In: ACM international conference proceeding series. https://doi.org/10.1145/2347635.2347643
- Rodgers PA (2018) Co-designing with people living with dementia. CoDesign. https://doi.org/10.1080/15710882.2017.1282527
- Smeenk W, Sturm J, Eggen B (2018) Empathic handover: how would you feel? Handing over dementia experiences and feelings in empathic co-design. CoDesign. https://doi.org/10.1080/ 15710882.2017.1301960
- Wu M, Richards B, Baecker R (2004) Participatory design with individuals who have amnesia. In: PDC 2004 - proceedings of the eight participatory design conference 2004 - artful integration:

interweaving media, materials and practices. https://doi.org/10. 1145/1011870.1011895

- 79. Sanders EBN, Dandavate U (1999) Design for experiencing: new tools. In: First international conference on design and emotion
- Sanders EB-N (2000) Generative tools for co-designing. In: Collaborative design. Springer, ???, pp 3–12. https://doi.org/10.1007/ 978-1-4471-0779-8\_1
- Bollard M, Magee P (2020) Myhealth-developing accessible health materials with men with intellectual disability: a cocreated proof of concept. Design Health. https://doi.org/10.1080/ 24735132.2020.1836862
- Brown DJ, Battersby S, Standen P, Anderton N, Harrison M, Brown DJ (2005) Inclusive design for disability: a prototype switch for use in virtual learning environments. Int J Disability Human Develop. https://doi.org/10.1515/IJDHD.2005.4.2.103
- Briggs-Goode A, Glazzard M, Walker S, Kettley S, Heinzel T, Lucas R (2016) Wellbeing and smart textiles: reflecting on collaborative practices and the design process. J Textile Design Res Practice. https://doi.org/10.1080/20511787.2016.1256599
- Baishya U, McLaren ES, Antle AN (2019) Idea bits: a tangible design tool to aid idea generation for tangible manipulation. In: Conference on human factors in computing systems - proceedings. https://doi.org/10.1145/3290607.3312820
- Hollinworth N, Kwiatkowska GM, Hwang F, Minnion A, Allen K (2014) Making electronics more accessible to people with learning disabilities. In: Conference on human factors in computing systems - proceedings. https://doi.org/10.1145/2559206.2581175
- Alfaras M, Primett W, Umair M, Windlin C, Karpashevich P, Chalabianloo N, Bowie D, Sas C, Sanches P, Höök K, Ersoy C, Gamboa H (2020) Biosensing and actuation-platforms coupling body input-output modalities for affective technologies. Sensors (Switzerland). https://doi.org/10.3390/s20215968
- Grove N, Walker M (1990) The Makaton vocabulary: using manual signs and graphic symbols to develop interpersonal communication. Augmentative Alternative Commun. https://doi.org/ 10.1080/07434619012331275284
- Liliana B, Filer B (2018) Evaluating cognitive accessibility. Scientific keys to strengthen the role of the evaluator with functional diversity (English edition). La Ciudad Accesible, ???
- Makhaeva J, Frauenberger C, Spiel K (2016) Creating creative spaces for co-designing with autistic children - the concept of a "Handlungsspielraum". In: ACM international conference proceeding series. https://doi.org/10.1145/2940299.2940306
- 90. Aswad E, Murphy E, Fernandez-Rivera C, Boland S (2022) Towards an inclusive co-design toolkit: perceptions and experiences of co-design stakeholders. Lecture notes in computer science (including subseries lecture notes in artificial intelligence and lecture notes in bioinformatics) 13342 LNCS, pp 284–292. https://doi.org/10.1007/978-3-031-08645-8\_33/COVER
- 91. Howard T, Venkatasubramanian K, Skorinko JLM, Bosma P, Mullaly J, Kelly B, Lloyd D, Wishart M, Alves E, Jutras N, Freark M, Alterio NA (2021) Designing an app to help individuals with intellectual and developmental disabilities to recognize abuse. ASSETS 2021 - 23rd international ACM SIGACCESS conference on computers and accessibility. https://doi.org/10.1145/3441852. 3471217
- 92. Sitbon L, Farhin S (2017) Co-designing interactive applications with adults with intellectual disability: a case study. ACM Int Conf Proc Series 487–491. https://doi.org/10.1145/3152771.3156163
- 93. Alkadhi B, Alnafisi G, Aljowair L, Alotaibi L, Alduaifi N, Alhumood R (2020) Co-design of augmented reality storybooks for children with autism spectrum disorder. Lecture notes in computer science (including subseries lecture notes in artificial intelligence and lecture notes in bioinformatics) 12426 LNCS, pp 3–13. https://doi.org/10.1007/978-3-030-60149-2\_1/COVER

- Cuadra A, Goedicke D, Zamfirescu-Pereira JD (2021) Democratizing design and fabrication using speech: exploring co-design with a voice assistant. ACM Int Conf Proc Series. https://doi.org/ 10.1145/3469595.3469624
- 95. Alves-Oliveira P, Paiva A, Arriaga P, Hoffman G (2017) Yolo, a robot for creativity: a co-design study with children. IDC 2017 - proceedings of the 2017 ACM conference on interaction design and children, pp 423–429. https://doi.org/10.1145/ 3078072.3084304
- Leahy C, Mcmahon K, Lloyd M, Bohemia P (2018) Exploring the pop-up shop for co-design research. DRS Biennial Conf Series 5:25–28. https://doi.org/10.21606/DRS.2018.457
- Guha ML, Druin A, Fails JA (2013) Cooperative inquiry revisited: reflections of the past and guidelines for the future of intergenerational co-design. Int J Child-Comput Interaction 1(1):14–23. https://doi.org/10.1016/J.IJCC1.2012.08.003
- Mohan HT, Nandanan K, Mohan R, Sadipe O, Williams I, Potocnik T (2019) Case study on co-design methodology for improved cook stove solutions for rural community in India. IEEE Region 10 humanitarian technology conference, R10-HTC. https://doi. org/10.1109/R10-HTC47129.2019.9042460
- 99. Wiethoff A, Schneider H, Rohs M, Butz A, Greenberg S (2012) Sketch-a-TUI: low cost prototyping of tangible interactions using cardboard and conductive ink. Proceedings of the 6th international conference on tangible, embedded and embodied interaction, TEI, pp 309–312. https://doi.org/10.1145/2148131.2148196
- Diewald S, Möller A, Roalter L, Kranz M (2014) Simulation and virtual prototyping of tangible user interfaces. Mensch und Computer 2014 - Tagungsband, pp 371–374. arXiv:1406.0370. https:// doi.org/10.1524/9783110344486.371
- Doherty G, Coyle D, Matthews M (2010) Design and evaluation guidelines for mental health technologies. Interacting Comput 22(4):243–252. https://doi.org/10.1016/j.intcom.2010.02.006
- 102. Jingar M, Lindgren H (2019) Tangible communication of emotions with a digital companion for managing stress: an exploratory co-design study. Proceedings of the 7th international Cconference on human-agent interaction. https://doi.org/10.1145/3349537
- 103. Søgaard Neilsen A, Wilson RL (2019) Combining e-mental health intervention development with human computer interaction (HCI) design to enhance technology-facilitated recovery for people with depression and/or anxiety conditions: an integrative literature review. Int J Mental Health Nursing 28(1):22–39. https://doi.org/ 10.1111/INM.12527
- 104. Mohiyeddini C, Semple S (2013) Displacement behaviour regulates the experience of stress in men. Stress
- 105. Hudson BF, Ogden J, Whiteley MS (2015) Randomized controlled trial to compare the effect of simple distraction interventions on pain and anxiety experienced during conscious surgery. European J Pain (United Kingdom)
- 106. Da Câmara SB, Agrawal R, Isbister K (2018) Identifying children's fidget object preferences: toward exploring the impacts of fidgeting and fidget-friendly tangibles. In: DIS 2018 - proceedings of the 2018 designing interactive systems conference. https://doi. org/10.1145/3196709.3196790
- Shibata T, Wada K (2011) Robot therapy: a new approach for mental healthcare of the elderly - a mini-review. Gerontology 57(4):378–386. https://doi.org/10.1159/000319015
- 108. Barker M, LindenJanet (2016) A sprite in the dark: supporting conventional mental healthcare practices with a tangible device. Proceedings of the tenth international conference on tangible, embedded, and embodied interaction - TEI '17
- 109. Balaam M, Fitzpatrick G, Good J, Luckin R (2009) Exploring affective technologies for the classroom with the subtle stone. Proceedings of the 28th international conference on human factors in computing systems - CHI '10, p 1623

- 110. Woodward K, Kanjo E, Umair M, Sas C (2019) Harnessing digital phenotyping to deliver real-time interventional bio-feedback. In: UbiComp/ISWC 2019- - adjunct proceedings of the 2019 ACM international joint conference on pervasive and ubiquitous computing and proceedings of the 2019 ACM international symposium on wearable computers. https://doi.org/10.1145/3341162. 3344838
- 111. Kröger JL, Lutz OHM, Raschke P (2020) Privacy implications of voice and speech analysis - information disclosure by inference. In: IFIP advances in information and communication technology. https://doi.org/10.1007/978-3-030-42504-3\_16
- Harold GT, Hay DF (2005) Normal development in middle childhood. Psychiatry 4(6):3–5. https://doi.org/10.1383/psyt.4.6. 3.66355
- 113. Saarni C (2011) Emotional development in childhood. Encyclopedia on Early Childhood Develpment
- 114. Bota PJ, Wang C, Fred ALN, Placido Da Silva H (2019) A review, current challenges, and future possibilities on emotion recognition using machine learning and physiological signals. https://doi.org/ 10.1109/ACCESS.2019.2944001
- Healey JA, Picard RW (2005) Detecting stress during real-world driving tasks using physiological sensors. IEEE Trans Intell Transport Syst 6(2):156–166
- 116. Woodward K, Kanjo E, Brown D, McGinnity TM (2020) Ondevice transfer learning for personalising psychological stress modelling using a convolutional neural network. In: On-device intelligence workshop, MLSys, Austin, Texas
- 117. Adams AT, Murnane EL, Adams P, Elfenbein M, Chang PF, Sannon S, Gay G, Choudhury T (2018) Keppi: a tangible user interface for self-reporting pain. Proceedings of the 2018 CHI conference

on human factors in computing systems - CHI '18. https://doi. org/10.1145/3173574.3174076

- Sharma N, Gedeon T (2012) Objective measures, sensors and computational techniques for stress recognition and classification: a survey. Comput Methods Programs Biomed 108(3):1287–1301
- 119. Schumm J, Zurich E, Ehlert U, Setz C, Arnrich B, Marca RL, Tröster G (2010) Discriminating stress from cognitive load using a wearable EDA device. IEEE Trans Inform Technol Biomed 14(2)
- 120. Alajmi N, Kanjo E, El Mawass N, Chamberlain A (2013) Shopmobia: an emotion-based shop rating system. In: 2013 Humaine association conference on affective computing and intelligent interaction. IEEE, ???, pp 745–750
- 121. Azevedo RT, Bennett N, Bilicki A, Hooper J, Markopoulou F, Tsakiris M (2017) The calming effect of a new wearable device during the anticipation of public speech. Sci Rep 7(1):2285
- 122. MacLean D, Roseway A, Czerwinski M (2013) MoodWings. In: Proceedings of the 6th international conference on pervasive technologies related to assistive environments - PETRA '13. ACM Press, New York, pp 1–8. https://doi.org/10.1145/2504335. 2504406
- 123. Gooch D, Mehta V, Price B, McCormick C, Bandara A, Bennaceur A, Bennasar M, Stuart A, Clare L, Levine M, Cohen J, Nuseibeh B (2020) How are you feeling? Using tangibles to log the emotions of older adults. TEI 2020 proceedings of the 14th international conference on tangible, embedded, and embodied interaction, pp 31–43. https://doi.org/10.1145/3374920.3374922

**Publisher's Note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.